



Integrating patient-centered factors in the risk assessment of MDS

Rena J. Buckstein

Odette Cancer Center, Sunnybrook Health Sciences Center, Toronto, ON, Canada

Myelodysplastic syndromes are clonal myeloid neoplasms that primarily present in older adults. Although leukemia develops in approximately 25% to 30% of individuals, the significantly shortened survival in this population is attributed more commonly to nonleukemic causes. The current prognostic scoring systems for leukemia and overall survival based on disease characteristics are becoming increasingly sophisticated and accurate with the incorporation of molecular data. The addition of patient-related factors such as comorbidity, disability, frailty, and fatigue to these new models may improve their predictive power for overall survival, treatment toxicity, and health care costs. To improve the generalizability of clinical trial results to the real world, geriatric assessment testing should become a standard of care in MDS clinical trials.

Learning Objectives

- Review the evidence that patient-related factors add independent value to current prognostic models
- Discuss the concept of frailty assessment and present several clinical tool options
- Envision a future of personalized prognostic scoring systems in MDS that are derived from big data sets using computer algorithms and the most important disease- and patient-related factors

Clinical case

A 70-year-old male retired English professor is referred with moderate pancytopenia. His laboratory values are hemoglobin 8.2 g/dL, platelets 98×10^9 /L, absolute neutrophil count 2.0×10^9 /L, mean corpuscular volume 107 fL, red cell distribution width 16%, lactate dehydrogenase 400 IU/L, and creatinine clearance 50 mL/min. He received his second transfusion 3 weeks ago. His medical history includes coronary artery disease with 2-vessel stenting 1 year ago for angina, type 2 diabetes on oral hypoglycemics, and a history of low-risk localized prostate cancer treated with prostatectomy >5 years ago without any biochemical/radiologic recurrence. He denies neuropathy or nephropathy. He does not exercise in any dedicated fashion other than routine walking. He is independent of all activities of daily living (ADLs) and instrumental activities of daily living (IADLs) but reports feeling fatigued or slowed, limiting his activities. You grade his Eastern Cooperative Oncology Group (ECOG) performance status as 1.

He is moderately overweight, with a body mass index of 30 kg/m². Current medications include metformin, aspirin, metoprolol, and rosuvastatin. A bone marrow aspirate and biopsy reveals dyserythropoesis and dysmegakarypoesis, blasts of 7%, and abnormal localization of immature precursors. His cytogenetics reveal trisomy 8,

and his next-generation sequencing results from a 40-gene panel reveal TET2 mutation (variant allele frequency 40%) and RUNX1 mutation (variant allele frequency 25%). He is diagnosed with myelodysplastic syndrome (MDS) with excess blasts type 1.

What is his prognosis?

Clinical prognostic scoring systems in MDS

There is no shortage of clinical prognostic scoring systems that incorporate disease-related characteristics in MDS primarily focusing on cytopenias, cytogenetics, bone marrow blast percentage, World Health Organization (WHO) classification, and molecular mutations.¹⁻⁴ They are used to predict leukemia-free survival (LFS) and overall survival (OS). Age is an additional significant prognostic factor, with MDS patients >75 years having a 40% greater risk of dying than younger individuals over a 27-month period,⁵ and it remains an independent risk factor incorporated into the international prognostic scoring system revised (IPSS-R) and the MD Anderson prognostic scoring system.^{3,4} According to the most commonly used scoring systems in clinical trials, our patient's disease has an IPSS score of 1.5 (high-intermediate, median OS 1.2 years) and an IPSS-R score of 6 (high risk, median OS 1.6 years). Factoring in his RUNX1 mutation, his prognosis is predicted to be slightly worse, ^{6,7} although the impact of comutations, variant allele frequency, and other clinical characteristics is still being resolved in an international effort⁶ without a "ready for use" deployable tool.

One critical limitation of these scoring systems for predicting OS is the primary focus on the disease (the seed) with neglect of the host factors (the soil) in which the disease has taken hold. If we are going to personalize our prognostic scoring systems and therapies based on dynamic disease biology, we must also personalize our predictive and prognostic scoring systems based on dynamic host biology.

Why is this important? MDS is predominantly a disease of the older adult, with a median age of presentation of 71 to 76 years, ^{8,9} with only

Conflict-of-interest disclosure: R.J.B. has received research funding from Celgene and Otsuka, and honoraria from Celgene. Off-label drug use: None disclosed.

Table 1. Comorbidities and their frequency in MDS

Study (N)	Cardiac	Vascular	Diabetes endocrine	Tumor	Pulmonary	Hepatic	GI	Psych	Renal	Any comorbidity, grades low, intermediate, high
Bammer et al ⁷⁴ (616)	25	7	12	10	5	3	6	2	2	48, 52, 25, 24
Naqvi et al ²⁵ (600)	CVS 54	_	16	28	9	_	7	8	2	77, 42, 21, 14
Della Porta et al ²⁶ (840 + 540)	25	5	16	10	3	17 (mild 14)	6	2	4	54, 65, 29, 6
Balleari et al ⁵ (318)	24	2	21	15	4	5	1	6	4	56,62,31,7
Zipperer et al ³⁰ (1161)	37	_	_	10	9	4	_	_	7	— , 50, 36, 14
Breccia et al ⁴⁰ (418)	38	6	18	6	6	5	6	2	5	93, 50, 25, 25
Falantes et al ⁷⁵ (232)	29	11	30	15	17	10	_	_	12	62, 44, 42, 14

CVS, cardiovascular system; GI, gastrointestinal; Psych, psychiatric.

6% cases diagnosed in those aged ≤50 years. An MDS diagnosis carries a significant increase in mortality that worsens with male sex, increasing age, and IPSS-R score. For our 70-year-old patient case, his survival is expected to be 43% that of age-matched controls in the United States. 10 This excess in mortality appears predominantly driven by factors unrelated to leukemic transformation. A large Spanish retrospective registry-based study documented a continued increase in excess mortality in recent years, despite the advent of hypomethylating agents and increased accessibility of allogeneic stem cell transplants. This might be ascribed to a parallel increase in age at diagnosis (from 70 in the years of 1980-1999 to 76 after 2006) and associated comorbidities.¹¹ In addition, predicting clinical outcomes for individual patients based on published trials in MDS may be erroneous, since older adults are underrepresented in clinical trials evaluating patients with hematologic cancers, 12,13 and patients in oncology trials tend to have better performance status and fewer comorbidities and remain on cytotoxic chemotherapy for longer durations.¹⁴

"Real-life" data demonstrate that the clinical outcomes of new therapies in MDS usually fall short of those from clinical trials, with median survivals in high-risk MDS patients treated with azacitidine (AZA) ranging from 12 to 16.4 months¹⁵⁻¹⁹ instead of the expected 24 months in the practice-changing AZA-001 clinical trial.²⁰

This is best exemplified by the Ontario, Canada experience with AZA. In an audit of 1101 patients with higher risk MDS/low-blast acute myelogenous leukemia (AML), the median survival was 11.6 months, and the median number of cycles received was 6 (instead of 9 cycles received on the AZA-001 trial). Ultimately, 30% of patients received <4 cycles of AZA with significantly inferior OS.²¹

Does the incorporation of patient-specific factors other than age into existing MDS risk assessment scores improve their performance?

Importance of comorbidity

Comorbidities increase with age, and a high prevalence of MDS patients have ≥1 comorbidities that either precede or follow their diagnosis. Commonly reported comorbidities and their frequencies are listed in Table 1 and are dominated by cardiac, endocrine, and other cancers. Comorbidities are significant, because they may affect therapeutic plans, tolerability, and outcomes in patients with cancer. They may also impact red blood cell transfusion thresholds, which are relevant to clinical trials in which transfusion independence is the primary end point. The independent impact of several comorbidity index scores have been evaluated in numerous, largely retrospective MDS studies summarized in Table 2. These include

the Charlson Comorbidity Index (CCI),23 the Hematopoietic Cell Transplantation-specific Comorbidity Index (HCT-CI), ²⁴ the Adult Comorbidity Evaluation-27 (ACE-27),²⁵ and the MDS-Comorbidity Index (MDS-CI).26 The independent contributions of CCI and HCT-CI to prognosis (for nonleukemic death and OS) have been demonstrated in different prognostic groups, making it difficult to determine which is the optimal comorbidity index to use. Some of the inconsistent findings are likely attributable to different disease and patient demographics, the use of disease-modifying treatments, and the imprecision of retrospective ascertainment. The most validated index is the MDS-CI. The MDS-CI derives from a large Italian cohort of MDS patients. Five comorbidities (cardiac, moderate to severe hepatic, severe pulmonary, renal, solid tumor) were found to be independently associated with the risk of nonleukemic death in multivariable analysis. A dynamic MDS-CI was developed composed of 3 risk groups (65% low, 29% intermediate, and 6% high) and predictive of OS and nonleukemic death independent of age, sex, WHO classification, cytogenetics, and transfusion dependency. Importantly, 32% had an increase of their MDS-CI over time, observed particularly in transfusion-dependent patients.²⁶ Our patient would score as high risk on the MDS-CI with his 3 comorbidities, but this may be less relevant to him, because the additional refinement of prognosis by the MDS-CI was primarily seen in patients with very low-, low-, and intermediate-risk WHO prognostic scoring system (WPSS) and IPSS-R⁵ scores. This suggests that the clinical relevance of mild or moderate comorbidity is trumped by the severity of MDS. In contrast, the comorbidities' refinement of prognosis in higher risk MDS in some studies might be attributed to their impact as competing causes of death, the restriction of therapeutic options or enrollment on clinical trials, and/or poor treatment tolerance.

Perhaps the severity of the comorbidity is more important than the actual tally. One comorbidity scale that evaluates severity is the ACE-27,²⁷ a validated index for patients with cancer that categorizes diseases into 1 of 3 levels of comorbidity with the overall comorbidity score based on the highest ranked and graded single ailment. In a retrospective cohort study of 600 patients at MD Anderson, 4 categories of comorbidity further refined the survival of patients with IPSS-R intermediate-, high-, and very high-risk patients. A final combined multivariable survival model and weighted risk score was derived, including age, ACE-27 score, and IPSS-R resulting in 4 risk groups with survivals of 53, 27, 13, and 7 months, respectively. No correlation was found between leukemic evolution and severity of comorbidity.²⁸ The patient discussed above has 4 comorbidities (coronary artery disease, diabetes, past cancer, and obesity), but all are graded mild; therefore, his ACE-27 score would be 1, and his

Table 2. Impact of comorbidities on OS

Study, year (N)	Prospective? Comorbidity scale(s)	MDS score	Predictive factors for OS/EFS by MVT analysis	Novel composite prognostic score derived?	Applicability/ comments	
Bammer et al, ⁷⁴ 2014 (616)	No, HCT-CI, CCI	IPSS	HCT-CI, IPSS, age	No	HCT-CI scores higher in men than women; more CVD in men	
Sperr et al, ⁷⁶ 2013 (400)	No, HCT-CI	IPSS	HCT-CI, ferritin >900, age	Yes		
Zipperer et al, ³⁰ 2008 (171)	No, HCT-CI, CCI	IPSS	HCT-CI, IPSS	No	Only in int-2 and high risk	
Naqvi et al, ²⁵ 2011 (600)	No, ACE-27	IPSS	ACE-27, age, IPSS	Yes	Only in int-1, int-2, and high-risk disease; patients ≤age 65	
Daver et al, ²⁹ 2014 (600)	No, ACE-27	IPSS-R	ACE-27, age, IPSS-R	Yes	Only in int-, high-, and very high-risk patients of all ages; same cohort from Naqvi et al ²⁶	
Della Porta et al, ²⁶ 2010 (840 derivation and 540 validation)	No, MDS-CI, CCI, HCT-CI	WPSS	MDS-CI, WPSS	No	Only in WPSS very low-, low-, and intermediate- risk disease	
Balleari et al, ⁵ 2015 (318)	No, MDS-CI, HCT-CI	IPSS, IPSS-R	MDS-CI, IPSS or IPSS- R, age >75, male	No	Only in IPSS/IPSS-R int- 2/int and high-, very high- risk patients	
van Spronsen et al, ²⁹ 2014 (222)	No, MDS-CI	IPSS, MDAS, WPSS, WPSSR, IPSSR	MDS-CI, IPSS-R, age, PS, fibrosis transfusions	No	Other MDS scoring systems were valid, but IPSS-R was best	
Zipperer et al, ³⁰ 2014 (1161)	No, MDS-CI	IPSS-R	MDS-CI, IPSS-R	No	Only when combined very low/low and intermediate-/high-/very high-risk	
Breccia et al, ⁷⁷ 2012 (418)	No, MDS-CI, CCI, HCT-CI	WPSS	CCI or MDS-CI, age >60, Hgb <8, Plt <50, bleeding	No	Higher MDS-CI and CCI scores associated with RBC transfusion dependency	
Falantes et al, ⁷⁵ 2016 (232)	No, MDS-CI	IPSS-R, IPSS	MDS-CI, age >75 y, diabetes on Tx, Hgb <10	Yes	Focus on NLD; patient population restricted to IPSS-low and int	
Buckstein et al, ⁶⁹ 2016 (445)	Yes, MDS-CI, CCI, HCT-CI	IPSS-R, IPSS	CCI frailty, IPSS	Yes	In separate models, MDS-CI was also prognostic, but the model with CCI had the largest R^2 ; frailty refined prognosis in all but IPSS-R very high risk	

CVD, cardiovascular disease; EFS, event-free survival; Hgb, hemoglobin; MVT, multivariate testing; NLD, nonleukemic death; Plt, platelet; PS, performance status; RBC, red blood cell; Tx, treatment.

composite score would be 7 (intermediate-2 [int-2]) with a predicted median survival of 13 months. Of note, this might still represent an underestimate of survival, since the combined survival model above was calculated from time of referral to MD Anderson and not diagnosis.²⁸ The MDS-CI has also been validated by other groups to add independent prognostic refinement to the IPSS-R.^{5,28-30} It is interesting to note that MDS-related causes of death declined in one study as the MDS-CI scores increased,³⁰ suggesting that optimizing life-limiting comorbidities such as cardiovascular disease might improve survival.

What is the best treatment for our patient?

Based on his higher risk IPSS/IPSS-R scores, a decision analyses³¹ and international guidelines that include the National Comprehensive Cancer Network³² and the European Leukemia network,³³ our patient should be treated with a hypomethylating agent with consideration of a

reduced-intensity allogeneic stem cell transplant. You recommend AZA to start and discuss the role of allogeneic stem cell transplant. Based on his hematopoietic stem cell transplant index (HCT-CI) score of 5, age, and high-risk disease, his projected 2-year non-relapse mortality is 39% following an allogeneic stem cell transplant. Based on this, many centers would discourage this approach in the absence of a clinical trial. With these statistics in hand, the patient is not interested in an allogeneic stem cell transplant but would consider a hypomethylating agent if it was well tolerated and could extend his life.

Will our case patient tolerate AZA, and what will be his expected OS?

Several tools that use data from a geriatric assessment, such as the Chemotherapy Risk Assessment Scale for High-Age Patients and the Cancer and Aging Research Group calculators, have been

developed to assist in the prediction of chemotherapy toxicity, but their derivation was primarily in solid tumor patients, so their applicability to MDS patients is unproven. ^{37,38}

Performance status and comorbidities also offer prognostic value for patients receiving AZA therapy. ECOG performance status ≥ 2 was found to be an independent component of the French prognostic score for OS developed in AZA treated patients with higher risk disease in addition to circulating blasts, transfusion dependency ≥ 4 units/8 weeks, and karyotype. ¹⁷ However, only a minority (20%) of patients in the French AZA registry fell into this poor performance status category. Because our patient's ECOG performance status is 1, he would fall into the French intermediate risk category (score 1-3) with a predicted median OS of 15 months. Performance status was also important at predicting outcome with AZA in the German PIAZA study. ECOG performance status (PS) of even 1 was associated with a progression-free survival that was 50% that of ECOG PS 0 (9.8 months vs 18.4 months, P = .002) in 150 patients with higher risk MDS, low-blast AML, or chronic myelomonocytic leukemia. ³⁹

Comorbidity may also impact treatment outcome with AZA. The MDS-CI has been shown to effectively risk stratify the OS from diagnosis of 60 patients treated with AZA (20 months for MDS-CI low risk, 12 months for MDS-CI intermediate risk, and 8 months for MDS-CI high risk (our patient).⁴⁰

With AZA treatment, do comorbidities increase the probability of unplanned emergency room visits or hospitalizations?

What additional health costs are borne by a public payer unrelated to AZA drug cost in patients with comorbidities (1 Canadian dollar = 0.74 US dollars in 2019)?

Comorbidity assessed using the Johns Hopkins Aggregated Diagnosis Groups (ADGs) may also be useful in predicting the tolerability and health care resource utilization associated with AZA. The ADG is a person-focused, diagnosis-based method of categorizing illnesses, often for large administrative database studies for predicting 1-year mortality in general ambulatory populations and health resource utilization. 41 Individual diseases or conditions are placed into a single ADG based on 5 clinical dimensions: duration of the condition, severity, diagnostic certainty, etiology, and specialty care involvement. The 32 possible ADGs are derived from both inpatient health administrative data as well as ambulatory health care data derived from physician billing claims.⁴² In an Ontario provincial registry using public administrative databases, we found that only transfusion dependence (hazard ratio [HR], 1.3; P = .0016) and a higher ADG score group of 10 to 12 (HR, 1.3; P = .0025) and 13+ (HR, 1.6; P < .0001) compared with 0 to 7 were associated with a higher number of emergency room visits, any hospitalization within the first 6 months of treatment, and a shorter time to hospitalization in 877 higher risk MDS/low-blast AML patients treated with AZA. A higher ADG comorbidity score (13+) was further predictive of increased health care spending (\$2029 above the base estimate of \$13 599 per cycle of AZA, excluding drug cost). 43

What are the limitations of using comorbidity and PS exclusively for "staging the aging"?

Poorer performance scores reliably identify patients at risk for more adverse outcomes; however, a minority (20% to 30%) of patients in MDS clinical trials and registries have Karnofsky performance scores of $<80^{44}$ or an ECOG PS that exceeds 1. Furthermore, 38% of

patients with a (physician-judged) PS of 0 to 1 have been found to have some disability on more careful geriatric testing. ⁴⁵ Comorbidity is a component but not a perfect surrogate of PS and "fitness." Not all patients with high comorbidity scores are "unfit," and not all unfit patients have high comorbidity scores.

What is the impact of patient-reported outcomes on survival?

The quality of life in MDS patients is inferior compared with agematched controls. The EL-NET investigators discovered the pronounced symptom burden experienced by many patients with MDS, predominantly in the dimensions of pain/discomfort, mobility, anxiety/ depression, and usual activities. ⁴⁶ Fatigue is one of the most commonly reported troublesome symptoms. ^{47,48} In addition to being a symptom that contributes negatively to quality of life, this patient reported outcome may, in and of itself have prognostic deterioration value for OS. In one study of higher risk MDS patients, every 10 points of fatigue (evaluated by the EORTC QLQ-C30), was associated with inferior survival (HR, 1.11-1.13) independent of the IPSS, IPSS-R, and WPSS risk scoring systems. 49 In a follow-up study that included validation in an independent cohort of higher risk MDS patients at Dana Farber, a fatigue score >45/100 provided the best result in distinguishing OS between patients with IPSS int-2 and high-risk disease. In fact, patients with IPSS int-2 and high fatigue and patients with IPSS high risk and low fatigue were found to have similar median OS of 15 to 16 months, allowing for the generation of a new fatigue-IPSS risk score in high-risk patients that reclassified them into 3 risk groups of 23 months, 16 months, and 10 months survival, respectively.⁵⁰

Are geriatric assessment tests feasible and do they add value in MDS?

The gold standard for detecting the older patient's vulnerability and frailty to poor outcome and treatment complications is the comprehensive geriatric assessment (CGA), which is strongly recommended by the International Society of Geriatric Oncology for cancer patients aged >70 years. 45 The CGA is multidimensional and encompasses the following domains: cognition, mood, ADLs, IADLs, mobility, polypharmacy, nutritional status, and social support. In a prospective study from Freiburg, Germany, Deschler et al found it feasible to administer 8 geriatric assessment instruments in 195 patients aged \geq 60 years with MDS (n = 63) or AML (n = 132). Patients with any impairment in ADL, a Karnofsky index <80%, and a fatigue score (European Organization for Research and Treatment of Cancer quality of life questionnaire [QLQ-C30]) of ≥50 had significantly worse OS than those treated with hypomethylating agents or best supportive care independent of cytogenetics, blasts, and comorbidity. 51 An Australian study demonstrated the feasibility of nurse-led "tailored" CGA in 98 patients with MDS and oligoblastic AML. Only 28% had ECOG scores of ≥2, yet 78% had ≥1 impairments in CGA domains indicating functional, cognitive, or psychological impairments. This culminated in 64% being referred to geriatricians for impactful "fine-tuning." Inability to complete IADLs and ADLs were independently poor prognostic factors of OS. IADL impairment in the 25 evaluable patients who received AZA was associated with significantly shorter survival (6 vs 19 months, P < .001), a higher failure rate to complete 6 cycles of AZA (71% vs 17%, P = .01), and a lower number of cycles of AZA (3.7 vs 12; P < .001). IADL impairment predicted a higher complication rate as well, with 50% ceasing treatment due to recurrent infections, prolonged hospital admissions, or poor tolerance compared with 17% ceasing treatment if they were IADL independent.⁵² In Canada, IADL dependence was also 1 of the 3 significant variables independently

predictive of OS in 239 lower risk, transfusion-dependent MDS patients in addition to IPSS-R and the receipt of iron chelation therapy. Its importance in the survival model is evidenced by an HR that matched that of IPSS-R higher risk (HR, 5; P = .0001). Fortunately, our patient has no ADL or IADL impairment to negatively impact his ability to complete an adequate trial of AZA.

Simple considerations such as nutrition/inflammation and walking ability may also be valuable. In a retrospective study from Dana Farber of 114 patients with largely higher risk MDS, ease at taking a long walk (no trouble vs some trouble; OS 53 vs 21 months) and low serum albumin (<3.5 g/dL; OS 4 vs 26 months) were found to be powerful independent prognostic factors of "equal impact" to the disease-related factors in a survival model.⁵⁴ The prognostic importance of a low serum albumin (<3.4 g/dL) for both OS and LFS was also demonstrated in a cohort of 200 MDS patients from Turkey.⁵⁵

What is the impact of frailty?

Frailty is defined as a medical syndrome with multiple causes and contributors that is characterized by diminished strength, endurance, and reduced physiologic function that increases an individual's vulnerability for developing increased dependency and/or death.⁵⁶ Cancer may contribute to or accelerate the frailty syndrome, and it may be an outcome of cancer treatment. Frailty has been associated with increased mortality in older patients with cancer, and its relevance in the management of hematologic malignancy, including MDS, has been excellently reviewed recently.⁵⁷ The 2 leading theories of frailty's pathophysiology include the frailty phenotype and the accumulated deficit theories.⁵⁸ In the frailty phenotype theory, frailty arises from aging-related cellular and physiological changes that lead to a condition of vulnerability. The components include 5 criteria: weight loss, low physical activity, weak grip strength, slow gait speed, and exhaustion.⁵⁹ The accumulated deficit theory postulates that vulnerability results from accumulated medical, physical, and social conditions that drive the increased vulnerability observed in frailty. 60 The concept is that a global system loses robustness as it develops various illnesses or functional declines, termed "deficits," and includes comorbidities and disability as deficits of age. An individual's frailty index is expressed as a ratio calculated as the number of deficits an individual has over the number of deficits considered in the index (usually a minimum of 30). It is scored as 0 to 1, with the lower end of the scale reflecting the robust and resilient and the upper end of the scale reflecting the vulnerable and frail.⁶¹

The best way to detect vulnerability and frailty is with a CGA. However, outside a clinical trial, a CGA may be too time consuming and resource intensive to deploy routinely, and there are a limited number of geriatricians.

Fortunately, there are available a number of quickly completed frailty screening tools endorsed by the International Society of Geriatric Oncology⁶² with the caveat that there is no "one tool fits all" that provides the optimal combination of sensitivity and specificity to identify older persons with an abnormal CGA.⁶³ Nevertheless, the consensus is that using a validated frailty tool such as the Vulnerable Elders Survey-13,⁶⁴ the Clinical Frailty Scale,⁶⁵ the modified Geriatric-8⁶⁶ (Figure 1) and others⁵⁸ may be useful to identify the vulnerable and frail at risk for adverse outcomes⁶⁷ and worthy of further assessment/intervention.

In one prospective study of patients with hematologic cancers that included MDS (23%) and AML (29%), an abnormal G8 (score of \leq 14/17) was an independent risk factor of mortality in addition to

a diagnosis of AML, impaired mobility, and risk of malnutrition. Of particular interest was the finding that the survival of patients with an abnormal G8 was short at 7.6 months irrespective of whether the patient received attenuated or no therapy versus standard therapy based on physician judgement.⁶⁸

The Rockwood Clinical Frailty Scale (CFS), which uses clinical descriptors and pictographs, was developed to provide clinicians with an easily deployable tool that stratifies older adults according to level of vulnerability and divides patients into 9 levels of vulnerability ranging from 1 (very fit) to 9 (terminally ill) (Figure 1). The CFS was validated in a sample of 2305 older participants from the Canadian Study of Health and Aging with each category increment of the scale increasing the medium-term risks of death and entry into an institution.⁶⁵ The MDS registry of Canada evaluated the impact of prospectively assessed frailty, comorbidity, disability and physical performance on OS and found that the incorporation of frailty (using the Rockwood scale) and comorbidity (CCI) improved the risk stratification of the IPSS-R by 30% and 5%, respectively, in a population of 445 MDS patients. A limitation of the derived composite prognostic score was the dichotomous definition of frailty (1-3 vs \geq 4) necessitated by the smaller sample sizes of patients falling into scores of ≥4 (25%).⁶⁹ While still impactful and informative, this prevented the more nuanced comparison of not frail (0-3) to vulnerable (4) to moderately frail (5-6) and severely frail (7-9) patients. Our patient would have been assigned a frailty score of 4, since he endorsed that symptoms limited his activities. In the Canadian composite risk score, he would have fallen into a high-risk group with a projected median survival of only 9.4 months. The Rockwood CFS has also been shown to add independent prognostic value to the IPSS-R and the CCI in a retrospective cohort of 118 patients diagnosed with MDS in a Japanese hospital. In this study, a CFS of ≥5 predicted a shorter 1-year OS (51% vs 90%, P < .001) and a higher incidence of infection-related mortality. 70 An MDS frailty index based on cumulative deficits has been developed by the Canadian MDS registry.71

Based on his CFS of 4 and MDS-CI of 4, what is the probability that our patient will remain on AZA for 4 or more cycles, and what is his expected survival?

In the Canadian MDS registry, the survival of the 63 out of 313 (20%) patients completing less than 4 cycles AZA was considerably less at 4 months (95% confidence interval [CI], 4-8) compared with 25 months (95% CI, 21-27) for \geq 4 cycles. A CFS of \geq 4 was also associated with shorter OS in the IPSS-R higher risk patients (10 vs 19 months, P = .04). The factors predictive of completing \geq 4 cycles included age, hemoglobin, platelet count, and MDS-CI, but by multivariable analysis, only MDS-CI was predictive. An MDS-CI score of \geq 2 was particularly discriminating (HR, 8.4; 95% CI, 2.3-33; P = .001). The factors predictive of the following predictive of the fol

In summary, our patient's CFS of 4 and high-risk MDS-CI would predict for a lower probability of completing 4 cycles of AZA and a survival of only 10 months if treated with AZA (instead of the predicted 19 months survival attached to his high IPSS-R score). This knowledge might impact the patient's decision and/or the physician's recommendation to pursue AZA treatment.

Conclusions and future directions

The seed and the soil are inextricably intertwined. The MDS prognostic scoring systems are effective for estimating LFS and OS and will be improved further with the inclusion of molecular data. The OS estimates could be improved by the inclusion of patient factors. This is

Rockwood Frailty Scale

Very Fit – People who are robust, active, energetic and motivated. These people commonly exercise regularly. They are among the fittest for their age.

 Well – People who have no active disease symptoms but are less fit than category 1. Often, they exercise or are very active occasionally, e.g. seasonally,

 Managing Well – People whose medical problems are well controlled, but are not regularly active beyond routine walking.



5 Mildly Frail – These people often have more wident slowing, and need help in high order IADLs (finances, transportation, heavy housework, medications). Typically, mild frailty progressively impairs shopping and walking outside alone, meal preparation and housework.

6 Moderately Frail – People need help with all the outside activities and with keeping house. Inside, the often have problems with stairs and need help with bathing and might need minimal assistance (cuing, standby) with dressing.

7 Severely Frail – Completely dependent for personal care, from whatever cause (physical or cognitive). Even so, they seem stable and not at high risk of dying (within ~ 6 months).

8 Very Severely Frail – Completely dependent, approaching the end of life. Typically, they could not recover even from a minor illness.



Note: Canadian Study on Health & Aging, Revised 2008. Rockwood K et al. A global clinical measure of fitness and frailty in elderly people. CMAJ 2005; 173:489-495. Copyright©2007-2009. Version 1.2. All rights reserved. Geriatric Medicine Research, Dalhousie University, Halifax, Canada. Permission granted to copy for research and educational purposes only.

Vulnerable Elders Survey – 13

	· - J
Items	Weight
1. Age	
- 75-84 years	1
-≥85 years	3
Self-evaluation of your	
health (Compared with	
other people your age)	
- Excellent	0
- Very good	0
- Good	0
- Fair	1
- Poor	ĺ
3. Do you have difficulty	Maximum 2 pts
on average, with the	Maximum 2 pts
following activities:	
a. Stopping, crouching,	1
or kneeling?	1
b. Lifting or carrying	1
objects as heavy as	1
5kg?	
c. Reaching or	1
extending arms above	1
shoulder level?	
d. Heavy housework	1
such as scrubbing	1
floors or washing	
windows?	
	1
e. Writing or handling and grasping small	1
objects?	
	1
f. Walking 500m? 4. Because of your health	M
	Maximum 4 pts.
or physical condition,	
do you have any	
difficulty?	4
a. Shopping for	4
personal items (like	
toilet items or	
medicines)?	4
b. Managing money? c. Walking across the	4
	4
room (use of a cane	
or walker is okay)?	4
d. Doing light	4
housework?	4
e. Bathing or	"
showering?	I

Note: Adapted from Saliba D, Elliott M, Rubenstein LZ, et al. The vulnerable elders survey: a tool for identifying vulnerable older people in the community. J Am Geriatr Soc. 2001;49(12):1691-1699. Copyright©2001, John Wiley and Sons.

*Score of \geq 3 is abnormal Maximum score: 10

Modified G8: G6 Frailty Score

G6 Score	Description	Weight
Weight loss average	≥ 3kg/does not	10
past 3 months	know	
	1-3 kg	2
	No weight loss	0
Neuropsychological	Mild/severe	3
problems	dementia or	
	depression	
	No problems	0
Polypharmacy	≥ 6 drugs	2
	< 6 drugs	0
Self-Rated Health	Poor/fair	3
Compared to other	Good/very good/	0
people of the same	excellent	
age, how does the		
patient rate his/her		
health		
Past history of heart	Congestive heart	5
failure or coronary	failure stroke or	
artery disease	myocardial	
	infarction	
	No History	0
Performance status	2-4	12
	1	4
	0	0

Note: Adapted from Martinez-Tapia C. Canoui-Poitrine F, Bastuji-Garin S, et al. Optimizing the G8 Screening Tool for Older Patients with Cancer: Diagnostic Performance and Validation of the Six-Item Version. The Oncologist 2016; 21: 188-195. Copyright©2016 AlphaMed Press.

*Score of > 6 is abnormal Maximum score: 35

Figure 1. (A) Geriatric Medicine Research, Dalhousie University, Halifax, Canada. Canadian Study on Health & Aging, Revised 2008. Version 1.2. Adapted from Rockwood et al⁶⁵ with permission. Permission granted to copy for research and educational purposes only. (B) Adapted from Saliba et al⁶⁴ with permission. (C) Adapted from Martinez-Tapia et al⁶⁶ with permission.

essential, since nonleukemia deaths predominate, the median age at diagnosis is increasing, and more therapies are emerging. Physicians are imperfect evaluators of "fitness." There is compelling evidence in MDS that supports some form of geriatric assessment at diagnosis and prior to the initiation of therapy. Screening geriatric tools may identify areas of vulnerability, predict toxicity and survival, facilitate health interventions, and help guide shared decision making. Many of these screening tools (like IADL, EQ-5D, fatigue, and frailty) are not time consuming, and some components can be completed by the patients at home or in the clinic waiting rooms. Geriatric screening tools

should be incorporated into all MDS clinical trials so that we can better discern (in parallel with more nuanced assessment of their methylome and genome) the patients who benefit from and tolerate therapy.

We have entered the era of big data and artificial intelligence. Last year, Nazha et al presented a more robust and validated personalized prediction model developed with artificial intelligence and machine learning algorithms from a training cohort of 1471 patients and validated in a cohort of 831 patients composed of 23 clinical variables (1 was age) inclusive of 11 somatic mutations with significantly better predictability than

existing prognostic scoring systems for OS and LFS.⁷³ It is probable that its predictability for OS could be further improved with the inclusion of some geriatric tests and/or comorbidities. Software programs that can independently extract diagnoses and laboratory values from clinical electronic patient records to create databases already exist. We should aspire toward an automated calculation of patients' frailty, comorbidity, and chemotherapy risk scores displayed in electronic patient records next to routine laboratory results and the derivation of personalized prognostic scores using the most impactful patient and disease elements.

References

- Greenberg P, Cox C, LeBeau MM, et al. International scoring system for evaluating prognosis in myelodysplastic syndromes. *Blood*. 1997;89(6): 2079-2088.
- Malcovati L, Germing U, Kuendgen A, et al. Time-dependent prognostic scoring system for predicting survival and leukemic evolution in myelodysplastic syndromes. *J Clin Oncol.* 2007;25(23):3503-3510.
- Kantarjian H, O'Brien S, Ravandi F, et al. Proposal for a new risk model in myelodysplastic syndrome that accounts for events not considered in the original International Prognostic Scoring System. *Cancer*. 2008; 113(6):1351-1361.
- Greenberg PL, Tuechler H, Schanz J, et al. Revised international prognostic scoring system for myelodysplastic syndromes. *Blood*. 2012; 120(12):2454-2465.
- Balleari E, Salvetti C, Del Corso L, et al. Age and comorbidities deeply impact on clinical outcome of patients with myelodysplastic syndromes. *Leuk Res.* 2015;39(8):846-852.
- Bejar R, Papaemmanuil E, Haferlach T, et al. Somatic mutations in MDS patients are associated with clinical features and predict prognosis independent of the IPSS-R: analysis of combined datasets from the International Working Group for Prognosis in MDS-Molecular Committee [abstract] *Blood.* 2015;126(23):907.
- Nazha A, Al-Issa K, Hamilton BK, et al. Adding molecular data to prognostic models can improve predictive power in treated patients with myelodysplastic syndromes. *Leukemia*. 2017;31(12):2848-2850.
- Ma X. Epidemiology of myelodysplastic syndromes. Am J Med. 2012; 125(7 suppl):S2-S5.
- Ma X, Does M, Raza A, Mayne ST. Myelodysplastic syndromes: incidence and survival in the United States. Cancer. 2007;109(8):1536-1542.
- Rollison DE, Howlader N, Smith MT, et al. Epidemiology of myelodysplastic syndromes and chronic myeloproliferative disorders in the United States, 2001-2004, using data from the NAACCR and SEER programs. *Blood*. 2008;112(1):45-52.
- Nomdedeu M, Pereira A, Ramos F, et al; Spanish MDS Group. Excess mortality in the myelodysplastic syndromes. Am J Hematol. 2017;92(2):149-154.
- Kanapuru B, Singh H, Myers A, et al. Enrollment of older adults in clinical trials evaluating patients with hematologic malignancies: the Food and Drug Administration (FDA) experience. *Blood.* 2017;130(suppl 1):861.
- Scher KS, Hurria A. Under-representation of older adults in cancer registration trials: known problem, little progress. J Clin Oncol. 2012;30(17):2036-2038.
- Anger C, Canta FAB, Mistry H, Mitrofan L, Karanis YB. 'Research' vs 'real world' patients: the representativeness of clinical trial participants. *Ann. Oncol.* 2016;27(suppl_6):1570P.
- Bernal T, Martínez-Camblor P, Sánchez-García J, et al; Spanish Society of Hematology. Effectiveness of azacitidine in unselected high-risk myelodysplastic syndromes: results from the Spanish registry. *Leukemia*. 2015;29(9):1875-1881.
- Zeidan AM, Sekeres MA, Garcia-Manero G, et al; MDS Clinical Research Consortium. Comparison of risk stratification tools in predicting outcomes of patients with higher-risk myelodysplastic syndromes treated with azanucleosides. *Leukemia*. 2016;30(3):649-657.
- Itzykson R, Thépot S, Quesnel B, et al; Groupe Francophone des Myelodysplasies(GFM). Prognostic factors for response and overall survival in 282 patients with higher-risk myelodysplastic syndromes treated with azacitidine. *Blood*. 2011;117(2):403-411.
- Zeidan AM, Stahl M, Sekeres MA, Steensma DP, Komrokji RS, Gore SD.
 A call for action: Increasing enrollment of untreated patients with higher-

- risk myelodysplastic syndromes in first-line clinical trials. *Cancer*. 2017; 123(19):3662-3672.
- Dinmohamed AG, van Norden Y, Visser O, et al. Effectiveness of azacitidine for the treatment of higher-risk myelodysplastic syndromes in daily practice: results from the Dutch population-based PHAROS MDS registry. *Leukemia*. 2015;29(12):2449-2451.
- Fenaux P, Mufti GJ, Santini V, et al. Azacitidine (AZA) treatment prolongs overall survival (OS) in higher-risk MDS patients compared with conventional care regimens (CCR): results of the AZA-001 phase III study [abstract]. *Blood.* 2007;110(11):817.
- Mozessohn L, Cheung MC, Fallahpour S, et al. Azacitidine in the 'real-world': an evaluation of 1101 higher-risk myelodysplastic syndrome/low blast count acute myeloid leukaemia patients in Ontario, Canada. Br J Haematol. 2018;181(6):803-815.
- Extermann M. Measurement and impact of comorbidity in older cancer patients. Crit Rev Oncol Hematol. 2000;35(3):181-200.
- Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. *J Chronic Dis.* 1987;40(5):373-383.
- Sorror ML, Maris MB, Storb R, et al. Hematopoietic cell transplantation (HCT)-specific comorbidity index: a new tool for risk assessment before allogeneic HCT. *Blood.* 2005;106(8):2912-2919.
- Naqvi K, Garcia-Manero G, Sardesai S, et al. Association of comorbidities with overall survival in myelodysplastic syndrome: development of a prognostic model. *J Clin Oncol*. 2011;29(16):2240-2246.
- Della Porta MG, Malcovati L, Strupp C, et al. Risk stratification based on both disease status and extra-hematologic comorbidities in patients with myelodysplastic syndrome. *Haematologica*. 2011;96(3):441-449.
- 27. Paleri V, Wight RG. Applicability of the adult comorbidity evaluation -27 and the Charlson indexes to assess comorbidity by notes extraction in a cohort of United Kingdom patients with head and neck cancer: a retrospective study. *J Laryngol Otol*. 2002;116(3):200-205.
- Daver N, Naqvi K, Jabbour E, et al. Impact of comorbidities by ACE-27 in the revised-IPSS for patients with myelodysplastic syndromes. Am J Hematol. 2014;89(5):509-516.
- van Spronsen MF, Ossenkoppele GJ, Holman R, van de Loosdrecht AA. Improved risk stratification by the integration of the revised international prognostic scoring system with the myelodysplastic syndromes comorbidity index. Eur J Cancer. 2014;50(18):3198-3205.
- Zipperer E, Tanha N, Strupp C, et al. The myelodysplastic syndromecomorbidity index provides additional prognostic information on patients stratified according to the revised international prognostic scoring system. *Haematologica*. 2014;99(3):e31-e32.
- Koreth J, Pidala J, Perez WS, et al. Role of reduced-intensity conditioning allogeneic hematopoietic stem-cell transplantation in older patients with de novo myelodysplastic syndromes: an international collaborative decision analysis. *J Clin Oncol.* 2013;31(21):2662-2670.
- Greenberg PL, Stone RM, Al-Kali A, et al. Myelodysplastic Syndromes, Version 2.2017, NCCN Clinical Practice Guidelines in Oncology. *J Natl Compr Canc Netw.* 2017;15(1):60-87.
- Malcovati L, Hellström-Lindberg E, Bowen D, et al; European Leukemia Net. Diagnosis and treatment of primary myelodysplastic syndromes in adults: recommendations from the European LeukemiaNet. *Blood*. 2013; 122(17):2943-2964.
- Sorror ML, Sandmaier BM, Storer BE, et al. Comorbidity and disease status based risk stratification of outcomes among patients with acute myeloid leukemia or myelodysplasia receiving allogeneic hematopoietic cell transplantation. J Clin Oncol. 2007;25(27):4246-4254.
- Sorror ML, Storb RF, Sandmaier BM, et al. Comorbidity-age index: a clinical measure of biologic age before allogeneic hematopoietic cell transplantation. J Clin Oncol. 2014;32(29):3249-3256.
- de Witte T, Bowen D, Robin M, et al. Allogeneic hematopoietic stem cell transplantation for MDS and CMML: recommendations from an international expert panel. *Blood*. 2017;129(13):1753-1762.
- Hurria A, Togawa K, Mohile SG, et al. Predicting chemotherapy toxicity in older adults with cancer: a prospective multicenter study. *J Clin Oncol*. 2011;29(25):3457-3465.

- Extermann M, Boler I, Reich RR, et al. Predicting the risk of chemotherapy toxicity in older patients: the Chemotherapy Risk Assessment Scale for High-Age Patients (CRASH) score. Cancer. 2012;118(13):3377-3386.
- Wehmeyer J, Zaiss M, Losem C, et al. Impact of performance status and transfusion dependency on outcome of patients with myelodysplastic syndrome, acute myeloid leukemia and chronic myelomonocytic leukemia treated with azacitidine (PIAZA study). Eur J Haematol. 2018;101(6):766-773.
- Breccia M, Salaroli A, Loglisci G, Finsinger P, Serrao A, Alimena G. MDSspecific comorbidity index is useful to identify myelodysplastic patients who can have better outcome with 5-azacitidine. *Haematologica*. 2012; 97(2):e2.
- Austin PC, van Walraven C, Wodchis WP, Newman A, Anderson GM.
 Using the Johns Hopkins Aggregated Diagnosis Groups (ADGs) to
 predict mortality in a general adult population cohort in Ontario, Canada.

 Med Care. 2011;49(10):932-939.
- Weiner JP, Abrams C, Bodycombe D. The Johns Hopkins ACG case-mix system version 6.0 release notes. Baltimore: Johns Hopkins Bloomberg School of Public Health; 2003.
- 43. Mozessohn L, Cheung M, Mittmann N, Earle CC, Liu N, Buckstein R. High healthcare utilization and costs in patients with higher-risk MDS/ low blast count AML treated with azacitidine in Ontario, Canada. *Blood*. 2018;132(suppl 1):834-834.
- 44. de Swart L, Smith A, Johnston TW, et al. Validation of the revised international prognostic scoring system (IPSS-R) in patients with lower-risk myelodysplastic syndromes: a report from the prospective European LeukaemiaNet MDS (EUMDS) registry. Br J Haematol. 2015;170(3):372-383.
- Wildiers H, Heeren P, Puts M, et al. International Society of Geriatric Oncology consensus on geriatric assessment in older patients with cancer. J Clin Oncol. 2014;32(24):2595-2603.
- Stauder R, Yu G, Koinig KA, et al. Health-related quality of life in lowerrisk MDS patients compared with age- and sex-matched reference populations: a European LeukemiaNet study. *Leukemia*. 2018;32(6):1380-1392.
- Steensma DP, Heptinstall KV, Johnson VM, et al. Common troublesome symptoms and their impact on quality of life in patients with myelodysplastic syndromes (MDS): results of a large internet-based survey. *Leuk Res.* 2008;32(5):691-698.
- Efficace F, Gaidano G, Breccia M, et al. Prevalence, severity and correlates of fatigue in newly diagnosed patients with myelodysplastic syndromes. Br J Haematol. 2015;168(3):361-370.
- Efficace F, Gaidano G, Breccia M, et al. Prognostic value of self-reported fatigue on overall survival in patients with myelodysplastic syndromes: a multicentre, prospective, observational, cohort study. *Lancet Oncol*. 2015;16(15):1506-1514.
- Efficace F, Cottone F, Abel G, et al. Patient-reported outcomes enhance the survival prediction of traditional disease risk classifications: an international study in patients with myelodysplastic syndromes. *Cancer*. 2018;124(6):1251-1259.
- Deschler B, Ihorst G, Platzbecker U, et al. Parameters detected by geriatric and quality of life assessment in 195 older patients with myelodysplastic syndromes and acute myeloid leukemia are highly predictive for outcome. *Haematologica*. 2013;98(2):208-216.
- Molga A, Wall M, Chhetri R, et al. Comprehensive geriatric assessment predicts azacitidine treatment duration and survival in older patients with myelodysplastic syndromes [published online ahead of print 22 April 2019]. J Geriatr Oncol. doi:10.1016/j.jgo.2019.02.002.
- 53. Leitch HA, Parmar A, Wells RA, et al. Overall survival in lower IPSS risk MDS by receipt of iron chelation therapy, adjusting for patient-related factors and measuring from time of first red blood cell transfusion dependence: an MDS-CAN analysis. Br J Haematol. 2017;179(1):83-97.
- 54. Fega KR, Abel GA, Motyckova G, et al. Non-hematologic predictors of mortality improve the prognostic value of the international prognostic scoring system for MDS in older adults. *J Geriatr Oncol.* 2015;6(4):288-298.
- Sevindik OG, Guc Z, Kahraman S, et al. Hypoalbuminemia is a surrogate biomarker of poor prognosis in myelodysplastic syndrome even when adjusting for comorbidities. *Leuk Lymphoma*. 2015;56(9):2552-2555.

- Morley JE, Vellas B, van Kan GA, et al. Frailty consensus: a call to action. J Am Med Dir Assoc. 2013;14(6):392-397.
- Abel GA, Klepin HD. Frailty and the management of hematologic malignancies. *Blood*. 2018;131(5):515-524.
- Huisingh-Scheetz M, Walston J. How should older adults with cancer be evaluated for frailty? J Geriatr Oncol. 2017;8(1):8-15.
- Fried LP, Tangen CM, Walston J, et al; Cardiovascular Health Study Collaborative Research Group. Frailty in older adults: evidence for a phenotype. J Gerontol A Biol Sci Med Sci. 2001;56(3):M146-M156.
- Rockwood K, Mitnitski A. Frailty in relation to the accumulation of deficits. J Gerontol A Biol Sci Med Sci. 2007;62(7):722-727.
- Song X, Mitnitski A, Rockwood K. Prevalence and 10-year outcomes of frailty in older adults in relation to deficit accumulation. *J Am Geriatr Soc.* 2010;58(4):681-687.
- Decoster L, Van Puyvelde K, Mohile S, et al. Screening tools for multidimensional health problems warranting a geriatric assessment in older cancer patients: an update on SIOG recommendations. *Ann Oncol*. 2015;26(2):288-300.
- 63. Hamaker ME, Jonker JM, de Rooij SE, Vos AG, Smorenburg CH, van Munster BC. Frailty screening methods for predicting outcome of a comprehensive geriatric assessment in elderly patients with cancer: a systematic review. *Lancet Oncol*. 2012;13(10):e437-e444.
- 64. Saliba D, Elliott M, Rubenstein LZ, et al. The Vulnerable Elders Survey: a tool for identifying vulnerable older people in the community. *J Am Geriatr Soc.* 2001;49(12):1691-1699.
- Rockwood K, Song X, MacKnight C, et al. A global clinical measure of fitness and frailty in elderly people. CMAJ. 2005;173(5):489-495.
- Martinez-Tapia C, Canoui-Poitrine F, Bastuji-Garin S, et al; ELCAPA Study Group. Optimizing the G8 screening tool for older patients with cancer: diagnostic performance and validation of a six-item version. Oncologist. 2016;21(2):188-195.
- 67. Turner G, Clegg ABritish Geriatrics SocietyAge UK; Royal College of General Practioners. Best practice guidelines for the management of frailty: a British Geriatrics Society, Age UK and Royal College of General Practitioners report. Age Ageing. 2014;43(6):744-747.
- Hamaker ME, Mitrovic M, Stauder R. The G8 screening tool detects relevant geriatric impairments and predicts survival in elderly patients with a haematological malignancy. *Ann Hematol.* 2014;93(6):1031-1040.
- Buckstein R, Wells RA, Zhu N, et al. Patient-related factors independently impact overall survival in patients with myelodysplastic syndromes: an MDS-CAN prospective study. Br J Haematol. 2016;174(1):88-101.
- Sakatoku K, Takeoka Y, Araki T, et al. Combination of frailty status and comorbidity score improves the prediction of survival in patients with myelodysplastic syndrome due to good predictive capability for infection-related mortality. *Blood*. 2016;128(22):1995.
- Starkman R, Wells RA, Chodirker L, et al. An MDS specific frailty index based on cumulative deficits adds independent prognostic value to established clinical prognostic scoring systems. *Blood.* 2017;130(suppl 1):1686.
- Blunt DN, Wells RA, Chodirker L, et al. Predictive factors of overall survival and the completion of four cycles of azacitidine: an MDS-CAN study. *Blood*. 2017;130(suppl 1):4253.
- Nazha A, Komrokji RS, Meggendorfer M, et al. A personalized prediction model to risk stratify patients with myelodysplastic syndromes. *Blood*. 2018;132(suppl 1):793.
- Bammer C, Sperr WR, Kemmler G, et al. Clustering of comorbidities is related to age and sex and impacts clinical outcome in myelodysplastic syndromes. *J Geriatr Oncol.* 2014;5(3):299-306.
- Falantes JF, Márquez-Malaver FJ, Knight T, et al. The incorporation of comorbidities in the prognostication of patients with lower-risk myelodysplastic syndrome. *Leuk Lymphoma*. 2017;58(8):1893-1902.
- Sperr WR, Kundi M, Wimazal F, et al. Proposed score for survival of patients with myelodysplastic syndromes. Eur J Clin Invest. 2013;43(11):1120-1128.
- Breccia M, Federico V, Latagliata R, et al. Evaluation of comorbidities at diagnosis predicts outcome in myelodysplastic syndrome patients. *Leuk Res.* 2011;35:159-162.